Author's response to reviews

Title: Validation of a questionnaire for assessing fecal impaction in the elderly: impact of cognitive impairment, and using a proxy

Authors:

Marta Barcelo (marta.barcelo@yahoo.es)
Maria Jose Jimenez-Cebrian (mjjimenez@valdeluz.com)
Manuel Díaz-Rubio (manueldiazrubio@gmail.com)
Alberto Lopez-Rocha (ROCHASOCIADO@terra.es)
Enrique Rey (rey.enrique@med.ucm.es)

Version: 3 Date: 1 April 2012

Author's response to reviews: see over
Dear Editor:

We submit the manuscript “Validation of a questionnaire for assessing fecal impaction in the elderly: impact of cognitive impairment, and using a proxy” to be considered by the editorial board for publication in BMC Geriatrics. It contains original data and is not under consideration elsewhere. The manuscript has been checked by a native English speaker.

It is signed by Marta Barcelo, Maria Jose Jimenez-Cebrian, Manuel Diaz-Rubio, Alberto Lopez Rocha and Enrique Rey. Authors do not have any financial conflicts to disclose.

We are looking forward to hearing from you soon.

Marta Barceló